From no-reply@babbage.sissa.it Tue May 18 10:12:12 1993
Received: from babbage.sissa.it by spica.npac.syr.edu (4.1/I-1.98K)
	id AA15016; Tue, 18 May 93 10:02:34 EDT
Message-Id: <9305181402.AA15016@spica.npac.syr.edu>
Received: by babbage.sissa.it
	(16.8/16.2) id AA23197; Tue, 18 May 93 16:03:37 +0200
Date: Tue, 18 May 93 16:03:37 +0200
Precedence: bulk
Note: bulletin board software written by PG at LANL (8/91, 12/92)
From: no-reply@babbage.sissa.it  (send mail ONLY to cond-mat)
Reply-To: cond-mat@babbage.sissa.it
To: paulc@npac.syr.edu
Subject: RE: get 9305019
Status: RO

%Paper: cond-mat/9305019
%From: "Hans Gerd Evertz, SCRI, (904) 644-0176" <EVERTZ@scri.fsu.edu>
%Date: Mon, 17 May 1993 16:32:49 -0400 (EDT)

%**********************************************************************
%  Vertex Models and Quantum Spin Systems: a nonlocal approach
%  H.G. Evertz, M. Marcu
%
%  LaTeX, no external figures.
%**********************************************************************
\documentstyle[11pt]{article}
\def\mybaselinestretch{1.0}   % normal 1.0; double spaced 1.5;
                              % This is not used on title page
%----------------------------------------------------------------------
\newif\ifdodraftheader
\newif\ifdomarginnotes
%         \dodraftheadertrue  % Comment away for final version !
%         \domarginnotestrue  % Comment away for final version !

%***********************************************************************
%   Information for title page: title, authors, abstract
%***********************************************************************
\def\Titleofthispaper{%
        Vertex Models and Quantum Spin Systems: \\
        A nonlocal approach%
     \footnote{To appear in
               {\em Computer Simulations in Condensed Matter Physics VI},
                ed.\ D.P. Landau, K.K. Mon, and H.B. Sch\"uttler,
                Springer Verlag, Heidelberg, Berlin, 1993.}
%
       \\[0.5em]
}%
   \def\Preprintnumber{\rule{0ex}{1ex}FSU-SCRI-93C-65  \\[.5ex]     %!!!
                                      cond-mat/9305019 }           %!!!
   \def\Preprintdate  {May 1993}
%
\def\Authorsofthispaper{
     {\bf Hans Gerd Evertz$^{1}$}
     {\bf and Mihai Marcu$^{2}$}\\[2em]}
\def\Theiraddresses{
     $^1\,$Supercomputer Computations Research Institute,        \\[-.2ex]
           Florida State University, Tallahassee, FL 32306    \\[-.2ex]
           evertz@scri.fsu.edu                                   \\[3ex]
     $^2\,$School of Physics and Astronomy,\\[-.2ex]
           Raymond and Beverly Sackler Faculty of Exact Sciences,\\[-.2ex]
           Tel Aviv University, 69978 Tel Aviv, Israel           \\[-.2ex]
           marcu@taunivm.bitnet  \\[3ex]   %\\[1cm]
}%
\def\Abstracttext{%
Within a general cluster framework, we discuss
the loop-algorithm, a new type of cluster algorithm that
reduces critical slowing down in vertex models and in quantum
spin systems.
We cover the example of the 6-vertex model in detail.
For the F-model, we present numerical results
that demonstrate the effectiveness of the loop algorithm.
We discuss how to
modify the original algorithm for
some more complicated situations, especially for
quantum spin systems in one and two dimensions.
%
%\\[2ex] \noindent { PACS} numbers:
%% (More appropriate keywords would be:)
%%  1)   ?         % Monte Carlo simulations: accelerated (Cluster) algorithms
%%  2)   ?         % Vertex models
%%  3)   75.10Jm   % Quantized Spin models
%%  4)   05.50+q,  % Lattice theory and statistics; Ising problems
%%
%        02.70+d,  % Computational Techniques
%        05.50+q,  % Lattice theory and statistics; Ising problems
%        75.10Jm,  % Quantized Spin models
%        68.35Rh   % Solid solid interfaces: phase trans. and crit. phenomena
%
}%% END OF ABSTRACTTEXT
%***********************************************************************
%
%-----------------------------------------------------------------------
%                   LAYOUT
%-----------------------------------------------------------------------
%\renewcommand{\baselinestretch}{1.5}          % for double spacing
%%%changed: ``\mybaselinestretch'' is switched-on after title page

%%! Using 12cm * 18cm   at 10pt to approximate Springer-Verlag requirements
\setlength{\textwidth}{15cm} %use 14cm to get all marginnotes
                             %on DinA4 completely
\setlength{\textheight}{21cm}
\setlength{\topmargin}{-10mm}
\setlength{\oddsidemargin}{5mm}
\setlength{\parskip}{1mm}
%
%\addtolength{\topmargin}{-1cm}
%\addtolength{\oddsidemargin}{-1cm}
%\addtolength{\textwidth}{2cm}
%\addtolength{\textheight}{2cm}
%
\setlength{\evensidemargin}{\oddsidemargin}

%-----------------------------------------------------------------------
    \newcommand{\note}[1]{}
    \ifdomarginnotes
                      \renewcommand{\note}[1]
                             %% {[{\bf #1\/}]\message{Note: #1}}
                             {\marginpar{\raggedright\tiny #1}
                              \message{Note: #1}}
              \fi
    \pagestyle{plain}
    \ifdodraftheader  \pagestyle{myheadings}
                  \markright{\today \hfil\hskip2ex \fbox{DRAFT} \hfil}
                  \addtolength{\headsep}{1em}
                  \setlength{\footskip}{0em}
             \fi
%
%-----------------------------------------------------------------------
%                 SECTIONING
%-----------------------------------------------------------------------
%Remark: here is how one can modify a TeX-command, using its old definition:
%        example:   \let\temp=\section
%                   \def\section#1{SECTION: \temp{#1}}

% In order to reduce the unpleasant large sizes of title in ``article'' style,
% we use \subsection    instead of \section
% and    \subsubsection instead of \subsection.

 \let\section=\subsection
 \let\subsection=\subsubsection
 \def\subsubsection#1{\subsection{#1}
      \par\note{CAUTION: subsection=subsubsection !}\par}
 \def\thesubsection{\arabic{subsection}}  % got rid of "\thesection."
 \let\thesection=\thesubsection    % in case some command uses
                                   % \thesection for labeling
%
%-----------------------------------------------------------------------
%                 ENVIRONMENTS
%-----------------------------------------------------------------------
%\newtheorem{definition}{Definition}

%-----------------------------------------------------------------------
%                 LABEL-COUNTERS
%-----------------------------------------------------------------------
%%% \renewcommand{\theequation}{{\protect\thesection.\arabic{equation}}}
%%% \renewcommand{\thetable}{{\protect{\bf \thesection.\arabic{table}}}}
%%% \renewcommand{\thefigure}{{\protect\bf\thesection.\arabic{figure}}}
%%% \newcommand{\mysection}[1]{\section{#1}
%%%                            \setcounter{equation}{0}
%%%                            \setcounter{figure}{0}
%%%                            \setcounter{table}{0}}
%
%%% IF you use the above \renewcommand{\theequation}... etc.,
%%% THEN you should also use \newcommand{\mysection}...
%%% in order to have suitable equation (etc.) numbering.
%%% THEN use "\mysection" instead of "\section" but use \section* as before

%-----------------------------------------------------------------------
%                 HYPHENATION
%-----------------------------------------------------------------------
 \hyphenation{ap-proxi-ma-tion}

%-----------------------------------------------------------------------
%                 NEW COMMANDS
%-----------------------------------------------------------------------
%
%.......................................................................
%                 Define \Time  (e.g. `20:26')
%.......................................................................
%
\newcount\hour   \newcount\minute
\hour  = \time   \divide\hour by 60
\minute= \hour   \multiply\minute by -60   \advance\minute by \time
\ifnum\minute<10 \def\Time{\number\hour:0\number\minute}
           \else \def\Time{\number\hour:\number\minute}  \fi

%.......................................................................
%                 Commands for generating SPACE
%.......................................................................
\newcommand{\figurebox}[2]{\fbox{\vbox to #1{\hbox to #2{\hfil}\vfil}}}
    				        % Empty placeholder for a figure
\newcommand{\stru}[1]{\rule{0ex}{#1}}   % for vspace in formulae and table
\newcommand{\str}{\rule{0ex}{2.7ex}}    % strut, to make a line higher
\newcommand{\tabhline}{\\[0.3ex] \hline \str}
\newcommand{\addcol}[1]{\hspace{2 \arraycolsep - #1}}
%
%.......................................................................
%                 List Making
%.......................................................................
\newcounter{itemnumber}
\newcommand{\mylist}[2]{\begin{list}{#1}{\usecounter{itemnumber}
                                         \setlength{\topsep}{0ex}
                                         \setlength{\itemsep}{0ex}
                                         \setlength{\parsep}{.3ex}
                                        }#2 \end{list}}
\newcommand{\aenumerate}[1]{\mylist{(\arabic{itemnumber})}{#1}}
\newcommand{\renumerate}[1]{\mylist{(\arabic{itemnumber})}{#1}}
\newcommand{\myitemize }[1]{\mylist{$-$                  }{#1}}
\let\mc=\multicolumn

%.......................................................................
%                 Common useful mathematical definitions
%.......................................................................
\newcommand{\Tr}{\mathop{\rm Tr}}	    % Trace
\newcommand{\Det}{\mathop{\rm Det}}	    % Determinant
\newcommand{\sgn}{\mathop{\rm sgn}}	    % Signum
\renewcommand{\Re}{\mathop{\rm Re}}	    % Real part
\renewcommand{\Im}{\mathop{\rm Ie}}	    % Imaginary part
%\newcommand{\ZZ}{\mathsym Z}                   % set of integer numbers
\newcommand{\scrZ}{{\cal Z}}                % script "Z"
\newcommand{\identity}{{\msym I}}	    % Identity matrix
\newcommand{\half}{{1\over2}}		    % One half
%
\newcommand{\mytimes}{\!\times\!}
\newcommand{\myequiv}{\!\equiv\!}
\newcommand{\myeq}{\!=\!}
\newcommand{\mylt}{\!<\!}
\newcommand{\mygt}{\!>\!}
\newcommand{\mygeq}{\!\geq\!}
\newcommand{\myleq}{\!\leq\!}
\newcommand{\myapprox}{\!\approx\!}
\newcommand{\mysp}{\mbox{~~~~~}}
\newcommand{\gsim}{\raisebox{-3pt}{$\stackrel{>}{\sim}$}}
\newcommand{\lsim}{\raisebox{-3pt}{$\stackrel{<}{\sim}$}}
%
%.......................................................................
%                 Some other useful abbreviations
%.......................................................................
\def\3{\ss}
\newcommand{\un}{\underline}
\newcommand{\subs}[1]{{\mbox{\scriptsize\it #1}}}
%
%.......................................................................
%                 Abbreviations for THIS PAPER
%.......................................................................
\newcommand{\YMP}{{\sc cray-ymp}}
%\newcommand{\tmin}{$t_{min}$ }
%
\def\cl{{\mbox{{\protect\scriptsize cl}}}}
\def\tot{{\mbox{{\protect\scriptsize tot}}}}
\def\kch{\half K_c}
\def\qmin{q_{\mbox{{\protect\scriptsize 3,min}}}}
\def\zcl{z^{\mbox{{\protect\scriptsize cl}}}}
\def\taucl{\tau^{\mbox{{\protect\scriptsize cl}}}}
\def\tauexp{\tau_{\mbox{{\protect\scriptsize exp}}}}
\def\tauint{\tau_{\mbox{{\protect\scriptsize int}}}}
\def\zexp{z_{\mbox{{\protect\scriptsize exp}}}}
\def\zint{z_{\mbox{{\protect\scriptsize int}}}}
\def\eminusK{\mbox{e}^{-K}}
\def\eplusK{\mbox{e}^K}
%
\def\zatKc{0.71(5)}
\def\zatKch{0.19(2)}
%
\newcommand{\U}{\tilde{u}}
\newcommand{\ovl}[1]{\overline{#1}}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%            MAKE TITLE PAGE (all text is defined at beginning of file)
%            No changes need to be made here.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\begin{document}
\begin{thebibliography}{9}  %:wq
%% \protect\small
%% \addtolength{\baselineskip}{-.1\baselineskip}
   \addtolength{\itemsep}{-\itemsep}
\def\mybf#1{#1} %boldface volume numbers look ugly
%
\bibitem{SW}
     R. H. Swendsen and J. S. Wang, Phys.\ Rev.\ Lett.\ 58 (1987) 86.
%
\bibitem{Wolff}
     R. C. Brower and P. Tamayo, Phys.\ Rev.\ Lett.\ 62 (1989) 1087;\\[1mm]
     U. Wolff, Phys.\ Rev.\ Lett.\ 62 (1989) 361,
               Nucl.\ Phys.\ B322 (1989) 759,
          and  Phys.\ Lett.\ 228B (1989) 379.
%
\bibitem{ClusterReviews}
     For reviews, see e.g.:\\[1mm]
     U. Wolff, in {\em Lattice '89\/}, Capri 1989, N. Cabbibo et al., editors,
     Nucl.\ Phys.\ B (Proc.\ Suppl.) 17 (1990) 93;\\[1mm]
%
     A. D. Sokal, in {\em Lattice '90\/}, Tallahassee 1990,
     U. M. Heller et al., editors,
     Nucl.\ Phys.\ B (Proc.\ Suppl.) 20 (1991) 55.
%
\bibitem{KandelDomany}
     D. Kandel and E. Domany, Phys.\ Rev.\ B43 (1991) 8539.
%
\bibitem{VMR}
H. G. Evertz, M. Hasenbusch, M. Marcu, K. Pinn and S. Solomon,
Phys.\ Lett.\ 254B (1991) 185, and
in {\em Workshop on Fermion Algorithms\/}, J\"ulich 1991,
H. J. Herrmann and F. Karsch editors,
Int.\ J. Mod.\ Phys.\ C3 (1992) 235.%(-246)
%
\bibitem{Z2gauge}
     R. Ben-Av, D. Kandel, E. Katznelson, P. Lauwers and S. Solomon,
     J. Stat.\ Phys.\ 58 (1990) 125. %\\[1mm]
     This cluster algorithm for the 3-dimensional $Z_2$ gauge theory
     may at first glance bear some resemblance to our loop algorithm,
     but the similarities are only superficial.
%
%\bibitem{FrustratedIsing}
%     D. Kandel, R. Ben-Av and E. Domany, Phys.\ Rev.\ Lett.\
%     65 (1990) 941.
%
\bibitem{BCSOScluster}
     M. Hasenbusch, G. Lana, M. Marcu and K. Pinn,
     {\it Cluster algorithm for a solid-on-solid model with constraints,}
%    Tel Aviv University preprint TAUP 1920-91,
     Phys.\ Rev.\ B46 (1992) 10472.
%
\bibitem{LoopPRL}
     H.G. Evertz, G. Lana and M. Marcu,
     Phys.\ Rev.\ Lett.\ 70 (1993) 875.
%
\bibitem{LoopSixVertex}
        % {\sl The Loop-Cluster Algorithm for the full 6-Vertex Model},
        H.G. Evertz and M. Marcu,
%       hep-lat/9211047, FSU-SCRI-92-165, November 1992,
%       to appear in {\em Lattice 92}, Amsterdam 1992, ed.\ J. Smit et al.,
        in {\em Lattice 92}, Amsterdam 1992, ed.\ J. Smit et al.,
        Nucl.\ Phys.\ B (Proc.\ Suppl.) 30 (1993) 277.
%
\bibitem{ReviewLieb}
    E. H. Lieb, Phys.\ Rev.\ Lett.\ 18 (1967) 1046;\\[1mm]
     E. H. Lieb and F. Y. Wu,
     {\em Two-dimensional Ferroelectric Models\/},
     in {\em Phase Transitions and Critical Phenomena\/} Vol.\ 1,
     C. Domb and M. S. Green, editors, (Academic, 1972) p.\ 331;\\[1mm]
%%%\bibitem{Baxter}
     R. J. Baxter, {\em Exactly Solved Models in Statistical
     Mechanics\/} (Academic, 1989).
%
\bibitem{QMC}
 For Quantum Monte Carlo simulations see:\\
     M. Suzuki editor,
     {\em Quantum Monte Carlo methods in equilibrium and nonequilibrium
     systems\/}, Taniguchi symposium, Springer (1987).
%
\bibitem{ParallelSW}
     See the overview in \cite{VectClust}.
%
\bibitem{VectClust}
%      {\sl Vectorized Search for Single Clusters},
       H.G. Evertz,
       J. Stat.\ Phys.\ 70 (1993) 1075.
%
%\bibitem{WieseYing}
%     U.J. Wiese, H.P. Ying,
%     %{\it Block spin cluster algorithms for quantum spin systems,}
%     %   Bern preprint BUTP-92-16.
%     Phys.\ Lett.\ 168A (1992) 143.
%
\bibitem{WieseHeisenberg}
        U.J. Wiese and H.P. Ying, Bern preprint;\\
        bulletin board cond-mat/9212006.
%
\bibitem{Signpaper}
        H.G. Evertz, in preparation.

\end{thebibliography}

%
% TOP OF TITLE PAGE: %%SPRINGER
  \parbox[t]{20ex}{\ifdodraftheader \fbox{\bf Draft version}   \fi}
  \hfill           \ifdomarginnotes \note{\today} \note{\Time} \fi
  \begin{tabular}[t]{l}
                     \rule{0ex}{1ex}\Preprintnumber \\[.5ex]
                     \rule{0ex}{1ex}\Preprintdate
  \end{tabular}
%
%\date{}
%\title{\vbox{\vspace{2ex}}  \Titleofthispaper}
%\author{\vbox{\vspace{3em}} \Authorsofthispaper}
%\begin{document}
%
% AUTHORS AND TITLE
    \renewcommand{\thefootnote}{{\protect\fnsymbol{footnote}}}
%        The following is adapted from article.sty (\maketitle)
     \vskip 2em
         \begin{center}
           {\Large \bf \vbox{\vspace{4ex}} \Titleofthispaper \par}\vskip1.5em
         {\normalsize \lineskip .5em
%%           {\large      \lineskip .5em
             \begin{tabular}[t]{c} \vbox{\vspace{3em}} %%SPRINGER3em}}
                                 \Authorsofthispaper
             \end{tabular}\par}
   {\normalsize \begin{tabular}[t]{c} \vbox{\vspace{0em}}
                                 \Theiraddresses
             \end{tabular}\par}  \vskip 1em
         \end{center} \par %% \vskip 1.5em
    \renewcommand{\thefootnote}{{\protect\arabic{footnote}}}
%
 \vfill
% ABSTRACT
 \pagebreak[3]
\begin{abstract} {\protect\normalsize
                   \noindent \Abstracttext}
\end{abstract}
%\vspace*{1em} %%SPRINGER
\vfill
%
%\pagestyle{empty}
\thispagestyle{empty}
\setcounter{page}{0}
\newpage
\renewcommand{\baselinestretch}{\mybaselinestretch} % for double spacing
\protect\small \protect\normalsize                  % do use \baselinestretch
%
%
%***********************************************************************
%        END OF DEFINITIONS: BEGIN MAIN TEXT OF PAPER
%***********************************************************************
%
%***********************************************************************
 \section{Introduction}
%***********************************************************************
%
For Monte Carlo simulations of many interesting physical situations,
critical slowing down is a major problem.
Standard simulation algorithms employ {\em local} update procedures
like the Metropolis and the heat bath algorithm.
With local updates, ``information'' travels slowly, like a random
walk. If the
relevant length scale is the correlation length $\xi$,
the number of updates necessary to decorrelate large regions,
i.e.\ the autocorrelation time $\tau$, grows like
%
\begin{equation}\label{CSD}
\tau \propto \xi^z,
\end{equation}
%
where $z \approx 2$ for local updates, as suggested by the
random walk analogy.

The way out of this problem is to employ $nonlocal$ updates.
The challenge is to devise algorithms that are nonlocal and still
satisfy detailed balance.
Multigrid algorithms are one possible approach.
In this paper we shall focus on {\em cluster algorithms};
for a nonexhaustive selection of references see
\cite{SW,Wolff,ClusterReviews,KandelDomany,VMR,Z2gauge}.

The first cluster algorithm was invented by
Swendsen and Wang (SW) \cite{SW} for the case of the
Ising spin model. The basic idea is to perform moves that significantly
change the Peierls contours characterizing a configuration.
As the size of Peierls contours is, typically, anything up to the
order of the correlation length, critical slowing down may be eliminated
completely or at least partially by this approach.
The SW algorithm has been modified and generalized for other spin systems,
mostly with two spin interactions \cite{Wolff,ClusterReviews,VMR}.
Notice that for these systems clusters are connected regions of spins,
with the same dimensionality as the underlying lattice.
A few generalizations along different lines were also done
\cite{KandelDomany,Z2gauge}.

Recently \cite{BCSOScluster,LoopPRL,LoopSixVertex} we introduced
{\em cluster algorithms for vertex models and quantum spin systems},
which are  the first cluster
algorithms for models with constraints. While \cite{BCSOScluster} is an
adaptation of the valleys-to mountains-reflections (VMR) algorithm
\cite{VMR}, originally devised for solid-on-solid models,
the {\em loop algorithm} introduced in \cite{LoopPRL,LoopSixVertex}
does not resemble any existing scheme.

In this paper we discuss the loop algorithm in detail.
In vertex models \cite{ReviewLieb} the dynamical variables
are localized on bonds, and the interaction is between all bonds meeting
at a vertex. Furthermore there are constraints on the possible bond
variable values around a vertex.
%
Our scheme is devised such as to take into account the constraints
automatically, and to allow a simple way to construct the  clusters.
%
The clusters here are not connected regions of spins,
but instead {\em closed paths} (loops) of bonds.

In what follows we first comment on the SW algorithm for spin systems.
Then we discuss the general cluster formalism of Kandel and Domany
\cite{KandelDomany}, treating the SW algorithm as an example.
Next we define the 6-vertex model, and, as a special case, the F~model.
We then introduce the loop algorithm, and show how to formulate it
for the complete 6-vertex model. The optimization of the algorithm is
discussed in a separate section. For the special case of the F-model,
we describe how the algorithm particularizes, and then we present
our very successful numerical tests of the loop algorithm.
We also show how to apply the loop algorithm to simulations of one and two
dimensional quantum spin systems \cite{QMC},
like e.g. the $xxz$ quantum spin chain,
and the spin $1\over2$ Heisenberg antiferromagnet and ferromagnet
in two dimensions.
Finally we comment on further generalizations and applications.

%***********************************************************************
 \section{Some comments on the Swendsen-Wang algorithm}
%***********************************************************************

Here we present a somewhat unusual viewpoint on the SW algorithm, in
order to better understand what is new in the loop algorithm.
Let us look at a spin system, like the Ising model,
with variables $s_x$ living on sites $x$,
and a nearest neighbor Hamiltonian $H(s_x,s_y)$.
The partition function of the model to be simulated is
$Z=\sum_{s} \exp{(-\sum_{<xy>} H(s_x,s_y))}$, where $<xy>$ is a pair
of sites.

Consider update proposals (flips) $s_x \rightarrow s'_x$,
such that $H(s_x,s_y) = H(s'_x,s'_y)$.
Then it follows that updating at the same time
all spins in some ``cluster'' of spins, we
will only change the value of the Hamiltonian at the {\em boundary} of that
cluster, not inside it.

In order to satisfy detailed balance,
we have to choose clusters with an appropriate probability.
The SW algorithm amounts to making a {\em Metropolis} decision
for each bond $<xy>$, namely whether to accept the change in $H$ from
$H(s_x,s_y)$ to $H(s'_x,s_y)$.
If accepted, a bond is called ``deleted'', otherwise it is called
``frozen''. Clusters are then sets of sites connected by frozen bonds.
Note that if deleted, a bond {\em may} be at the boundary of a cluster,
but need not.

Finally, an update is performed by finding all the clusters in a given
configuration and then flipping each cluster with 50\% probability
\cite{SW}. In Wolff's single cluster variant \cite{ClusterReviews,Wolff},
which is dynamically more favourable,
we construct one cluster starting from a randomly chosen
initial site, and then flip it with 100\% probability.

A technical remark: The Swendsen-Wang algorithm can be
vectorized and parallelized \cite{ParallelSW}.
The difficult task is to identify
the clusters, which is the same task as e.g.\ image component labeling.
For the single cluster variant, vectorization is the most efficient
approach \cite{VectClust}.

%***********************************************************************
 \section{The Kandel - Domany framework.}
%***********************************************************************

Cluster algorithms are conveniently described in the general framework
of Kandel and Domany \cite{KandelDomany}.
%Actually, this is a framework for stochastically mapping between different
%models, while preserving detailed balance.
Let us consider the partition function
%
\begin{equation}\label{ZKandel}
Z=\sum_{u} \exp{(-V(u))},
\end{equation}
%
where $u$ are the configurations to be summed over. We shall call
the function $V$ the ``interaction''.
Let us also define a {\em set} of new interactions $\ovl{V}_i$
(the index $i$ numbers the modified interactions).
Assume that during a Monte Carlo simulation we arrived at a given
configuration $u$. We choose a new configuration in a two step
procedure. The first step is to replace $V$ with one of the
$\ovl{V}_i$. For a given $i$, $\ovl{V}_i$ is chosen with
probability $p_i(u)$. The $p_i(u)$ satisfy:
%
\begin{equation} \label{KDconstraints}
\begin{array}{rll}
       p_i(u) &=& exp{(V(u) -  \ovl{V}_i(u) +c_i)}, \\[0.5ex]
\sum_i p_i(u) &=& 1 \; ,
\end{array}
\end{equation}
%
where $c_i$ are constants independent of $u$.
The second step is to update the configuration by employing a
procedure that satisfies detailed balance for the chosen $\ovl{V}_i$.
The combined procedure satisfies detailed balance for the
original interaction $V$ \cite{KandelDomany}.

In many cases, the interaction $V$ is a sum over local functions $H^c$,
where $c$ typically are cells of the lattice (like bonds in the
spin system case, sites for vertex models, plaquettes for gauge theories).
More generally, $H^c$ can contain part (or all) of the interactions in
some neighborhood of the cell $c$.
We can choose the modified interactions
$\ovl{H^c}_{i^c}$ for each $H^c$ independently ($i^c$ numbers the
possible new interactions for the cell $c$). The probabilities
for this choice obey eq.\ (\ref{KDconstraints}),
with $V$ replaced by
$H^c$, and $i$ by $i^c$. The total modified interaction
$\ovl{V}_i$ will then be the sum of all the $\ovl{H^c}_{i^c}$
($i$ is now a multiindex). When clear from the context, we shall
drop the index $c$.

Take the Ising model as an example.
The configuration $u$ is comprised of spins
$s_x = \pm 1$, the cells where the interaction is localized are
bonds $<xy>$, and $V(u) = -J \sum_{<xy>} s_x s_y$. We choose the bonds
as our cells $c$, so we can perform the first step of the
Kandel-Domany procedure separately for each bond.
The original cell interaction is
%
\begin{equation}
H^{<xy>}(s) = -J s_x s_y \;.
\end{equation}
%
Now, let us define two new bond interactions; the first ($i^{<xy>}=1$)
is called ``freeze'', the second ($i^{<xy>}=2$) is called ``delete'':
%
\begin{equation}\label{KDising}
%\begin{array}{lll}
%\ovl{H^{<xy>}}_{\mbox{\small delete}}(s) &=& 0 \; ; \\[1ex]
%\ovl{H^{<xy>}}_{\mbox{\small freeze}}(s) &=&
%   \left\{ \begin{array}{ll}  0,  & s_x = s_y \\
%            \infty, & else   \end{array}  \right. .
%\end{array}
%
\ovl{H^{<xy>}}_{\mbox{\small freeze}}(s) =
   \left\{ \begin{array}{ll}  0,  & s_x = s_y \\
            \infty, & else   \end{array}  \right. ; \quad\quad
\ovl{H^{<xy>}}_{\mbox{\small delete}}(s) = 0 \; .
%
\end{equation}
%
Then from (\ref{KDconstraints})
we obtain the Swendsen-Wang probabilities
if we choose the constants $c_i$ that minimize freezing:
%
\begin{equation}
\begin{array}{lll}
p^{<xy>}_{\mbox{\small delete}}(s) &=&
  \exp{(-J (s_x s_y + 1))} \, = \, \min{(1,\exp{(-2J s_x s_y)})} \;, \\[1.5ex]
%
p^{<xy>}_{\mbox{\small freeze}}(s) &=&
  1 - p^{<xy>}_{\mbox{\small delete}}(s) \;,
\end{array}
\end{equation}
%
which is just the Swendsen-Wang probability.

%***********************************************************************
 \section{The 6-Vertex Model}
%***********************************************************************

The 6-vertex model \cite{ReviewLieb} is defined on a square
lattice. On each bond there is an Ising-like variable that is usually
represented as an arrow. For example, arrow up or right means $+1$,
arrow down or left means $-1$. At each vertex we have the constraint
that there are two incoming and two outgoing arrows.
In fig.~1
%\ref{figsixvertices}
we show the six possible configurations
at a vertex, numbered as in \cite{ReviewLieb}.
The statistical weight of a configuration is given by the product over
all vertices of the vertex weights $\rho(u)$. Thus a priori,
for each vertex there are 6 possible weights $\rho(u)$, $u=1,...,6$.
We take  the weights to be symmetric under
reversal of all arrows. Thus, in standard notation
\cite{ReviewLieb}, we have:
\begin{equation}\label{sixvertexweights}
\begin{array}{l}
  \rho(1)  =  \rho(2)  =  a \; , \\
  \rho(3)  =  \rho(4)  =  b \; , \\
  \rho(5)  =  \rho(6)  =  c \; .
\end{array}
\end{equation}
The 6-vertex model has two types of phase transitions: of
Kosterlitz-Thouless type and of KDP type \cite{ReviewLieb}.
A submodel exhibiting the former is the F~model, defined by
$c=1$, $a=b=\exp{(-K)}$, $K\geq0$.
For the latter transition an example is the KDP
model itself, defined by $a=1$, $b=c=\exp{(-K)}$, $K\geq0$.

%%%%%%% PICTURE OF VERTICES %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{figure}[tb] \label{figsixvertices}
\begin{center}
%
\setlength{\unitlength}{.0005\textwidth}
%\begin{picture}(1700,300)(0,20)  % would be "correct"
\begin{picture}(1700,200)(0,50) % 200: higher; 90: caption closer
\thicklines
%
%%% DEFINE ARROWS (length 100) and VERTEX
\newsavebox{\RIGHT}
\newsavebox{\LEFT}
\newsavebox{\UP}
\newsavebox{\DOWN}
\sbox{\RIGHT}{\put(  0,0  ){\vector( 1, 0){70}}\put(70,0 ){\line(1,0){30}}}
\sbox{\UP}   {\put(  0,0  ){\vector( 0, 1){70}}\put(0 ,70){\line(0,1){30}}}
\sbox{\LEFT} {\put(100,0  ){\vector(-1, 0){70}}\put(0 ,0 ){\line(1,0){30}}}
\sbox{\DOWN} {\put(  0,100){\vector( 0,-1){70}}\put(0 ,0 ){\line(0,1){30}}}
\newcommand{\VERTEX}[5]{\begin{picture}(0,0)
                              \put(-100,0   ){\usebox{#1}}       %left
                              \put(   0,0   ){\usebox{#2}}       %right
                              \put(   0,-100){\usebox{#3}}       %bottom
                              \put(   0,0   ){\usebox{#4}}       %top
                              \put(   0,-170){\makebox(0,0){#5}} %text
                        \end{picture}}
%%% NOW DRAW 6 VERTICES
%
\put( 100,200){\VERTEX{\RIGHT}{\RIGHT}{\UP}  {\UP}  {1}}
\put( 400,200){\VERTEX{\LEFT} {\LEFT} {\DOWN}{\DOWN}{2}}
\put( 700,200){\VERTEX{\RIGHT}{\RIGHT}{\DOWN}{\DOWN}{3}}
\put(1000,200){\VERTEX{\LEFT} {\LEFT} {\UP}  {\UP}  {4}}
\put(1300,200){\VERTEX{\RIGHT}{\LEFT} {\DOWN}{\UP}  {5}}
\put(1600,200){\VERTEX{\LEFT} {\RIGHT}{\UP}  {\DOWN}{6}}
%
\end{picture}
%
\caption[fig1]{\parbox[t]{.80\textwidth}{
               The six vertex configurations, $u=1,...,6$
               (using the standard conventions of \cite{ReviewLieb}).}}
\end{center}
\vskip-1.5ex   % less space between figure and text
\end{figure}
%%%%%%%%%%%%%% END OF PICTURE %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%



\section{The Loop Algorithm}

If we regard the arrows on bonds as a vector field, the constraint at
the vertices is a {\em zero-divergence condition}. Therefore every
configuration change can be obtained as a sequence of {\em loop-flips}.
%operations defined as follows.
By ``loop'' we denote an oriented, closed,
non-branching (but possibly self-intersecting)
path of bonds, such that all arrows along the path point in the
direction of the path. A loop-flip reverses the direction  of
%(``flips'')
all arrows along the loop.

Our cluster algorithm performs precisely such operations, with
appropriate probabilities.
It constructs closed paths consisting of one or
several loops without common bonds. All loops in this path are flipped
together.

We shall construct the path iteratively, following the direction of the
arrows. Let the bond $b$ be the latest addition to the path. The arrow on
$b$ points to a new vertex $v$. There are two outgoing arrows at $v$,
and what we need is a unique prescription for continuing the path
through $v$. This is provided by a {\em break-up} of the vertex $v$.
In addition to the break-up, we have to allow for {\em freezing} of $v$.
By choosing suitable probabilities for break-up and freezing we shall
satisfy detailed balance.

The {\em break-up} operation is defined by splitting $v$ into two
pieces, as shown in fig.~2.
%\ref{figbreakup}.
The two pieces are either two
corners or two straight lines. On each piece, one of the arrows points
towards $v$, while the other one points away from $v$.
Thus we will not allow, e.g., the ul--lr
break-up for a vertex in the configuration 3.
If we break up $v$, the possible new configurations are obtained by
flipping (i.e., reversing both arrows of) the two pieces independently.
On the other hand, if we freeze $v$, the only possible configuration
change is to flip all four arrows.

%
%%%%%%% PICTURE OF BREAK-UP OF VERTICES AND OF STRAIGHT LINES %%%%%%%%%%%%
\begin{figure}[tb] \label{figbreakup}
\begin{center}
%
%%%\setlength{\unitlength}{.0005\textwidth}
\setlength{\unitlength}{.0004\textwidth} % to fit in one column
%\begin{picture}(1020,310)(0,20) % would be "correct"
\begin{picture}(1020,210)(0,50) % 260: move higher; 80: caption closer
\thicklines
%
\put(100,190){\line(-1, 0){100}}
\put(100,190){\line( 0,-1){100}}
\put(120,210){\line( 1, 0){100}}
\put(120,210){\line( 0, 1){100}}
\put(115,20){\makebox(0,0){ll--ur}}
%
\put(500,210){\line(-1, 0){100}}
\put(500,210){\line( 0, 1){100}}
\put(520,190){\line( 1, 0){100}}
\put(520,190){\line( 0,-1){100}}
\put(510,20){\makebox(0,0){ul--lr}}
%
\put(910,200){\line(-1, 0){110}}  % left
\put(910,200){\line( 1, 0){110}}  % right
\put(910,180){\line( 0,-1){ 90}}  % down
\put(910,220){\line( 0, 1){ 90}}  % up
\put(910,200){\oval(40,40)[l]}    % half-circle, left
\put(910,20){\makebox(0,0){straight}}
%
\end{picture}
\vspace*{1ex}
\caption[fig2]{ \parbox[t]{.80\hsize}{
                The three break-ups of a vertex:
                ll--ur (lower-left--upper-right),
                ul--lr (upper-left--lower-right),
                and straight.
              } }
\end{center}
\vskip-1.5ex   % less space between figure and text
\end{figure}
%
%%%%%%%%%%%%% END OF PICTURE %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%

The break-up and freeze probabilities are conveniently described within
the Kandel Domany framework.
%the general framework for cluster algorithms proposed by Kandel and
%Domany \cite{KandelDomany}.
It is sufficient to give them for one
vertex, which is in the current configuration $u$.
We define 6 new interactions (weight functions) $\rho_i$, $i=1,...,6$,
corresponding to specific break-up and freeze operations
(the labelling of the new interactions is completely
arbitrary, and the fact that we have six of them is just a coincidence).
For each vertex in configuration $u$, we replace
with probability $p_i(u)$ the original interaction $\rho$ by
the new interaction $\rho_i$.
Equation \ref{KDconstraints}
, i.e.\
detailed balance and the proper normalization of probabilities,
require that for every $u$
\begin{equation}\label{KDforsixvertex}
p_i(u) = q_i \frac{\rho_i(u)}{\rho(u)} \; , \quad \sum_i p_i(u) = 1 \; ,
\end{equation}
where $q_i =\exp{(-c_i)} \geq 0$ are parameters.

%%%%%%%%%%%% TABLE : newinteractions  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{table}[tb]    \label{tabnewinteractions}
\centering
\begin{tabular}{|c|c|c|c|}
\hline\str
 $i$ & action & $\rho_i(\U)$ & $p_i(u)$
\\[.5ex]\hline \stru{4ex}
1 & freeze 1,2 & \parbox[c]{20mm}{1, $\rho(\U)\!=\!a$ \\ 0, else}
               & \parbox[c]{21mm}{$q_1/a$, $\rho(u)\!=\!a$ \\ 0, else}
  \\[2.0ex]\hline \stru{4ex}
2 & freeze 3,4 & \parbox[c]{20mm}{1, $\rho(\U)\!=\!b$ \\ 0, else}
               & \parbox[c]{20mm}{$q_2/b$, $\rho(u)\!=\!b$ \\ 0, else}
  \\[2.0ex]\hline \stru{4ex}
3 & freeze 5,6 & \parbox[c]{20mm}{1, $\rho(\U)\!=\!c$ \\ 0, else}
               & \parbox[c]{20mm}{$q_3/c$, $\rho(u)\!=\!c$ \\ 0, else}
  \\[2.0ex]\hline \stru{4ex}
4 & ll--ur     & \parbox[c]{20mm}{0, $\rho(\U)\!=\!a$ \\ 1, else}
               & \parbox[c]{20mm}{0, $\rho(u)\!=\!a$ \\ $q_4/\rho(u)$, else}
  \\[2.0ex]\hline \stru{4ex}
5 & ul--lr     & \parbox[c]{20mm}{0, $\rho(\U)\!=\!b$ \\ 1, else}
               & \parbox[c]{20mm}{0, $\rho(u)\!=\!b$ \\ $q_5/\rho(u)$, else}
  \\[2.0ex]\hline \stru{4ex}
6 & straight   & \parbox[c]{20mm}{0, $\rho(\U)\!=\!c$ \\ 1, else}
               & \parbox[c]{20mm}{0, $\rho(u)\!=\!c$ \\ $q_6/\rho(u)$, else}
\\[2.0ex] \hline
\end{tabular}
%
\caption[dummy]{ \parbox[t]{0.85\textwidth}{
                  The new interactions $\rho_i(\U)$
                  and the probabilities $p_i(u)$
                  to choose them at a vertex in current configuration~$u$.
                  See eq.\ (\ref{KDforsixvertex}).
                }}
%
\end{table}
%
As discussed in \cite{KandelDomany}
(see also table~1),
%\ref{tabnewinteractions}),
{\em freezing} is described by introducing one new interaction for each
different value of $\rho(u)$.
For example, to freeze the value $a$, we choose the
interaction $\rho_1$ to be
$\rho_1(\U)=1$ if $\rho(\U)=a$, and $\rho_1(\U)=0$ otherwise.
In other words,
%%if $\U$ is 1 or 2, the Boltzmann weight $\rho_1(\U)$ is one, so
when $\rho_1$ is chosen,
transitions between $\U=1$ and $\U=2$ cost nothing,
whereas the vertex configurations 3, 4, 5, and 6 are
%however not allowed with $\rho_1$.
then not allowed.
Notice that we denote by $u$ the current configuration, and by $\U$
the argument of the function $\rho_i$.

Each {\em break-up} is also described by one new interaction. As an example
take the ul--lr break-up. It is given by the new interaction number five,
with $\rho_5(\U) = 1$ if $\rho(\U)=a$ or $c$,
and $\rho_5(\U) = 0$ if $\rho(\U)=b$.
In other words, with the new interaction $\rho_5$, transitions
between 1, 2, 5 and 6 cost nothing, while the vertex configurations 3
and 4 are not allowed.
This corresponds precisely to allowing independent corner flips
in a ul--lr break-up (see figs.~1,2).
%\ref{figsixvertices},\ref{figbreakup}).

Table~1 gives the full list of new weights $\rho_i(\U)$ and probabilities
$p_i(u)$
%%(the latter depend on the current configuration $u$, not on the
%%argument $\U$ of $\rho_i$)
to choose them in current configuration $u$.
%\ref{tabnewinteractions}.
{}From (\ref{KDforsixvertex}) we also obtain:
\begin{equation}\label{KDnorm}
\begin{array}{l}
q_1+q_5+q_6=a \; , \\
q_2+q_4+q_6=b \; , \\
q_3+q_4+q_5=c \; .
\end{array}
\end{equation}

Assume now that we have broken or frozen all vertices. Starting from a
bond $b_0$, we proceed to construct a closed path by moving in the arrow
direction. As we move from vertex to vertex, we always have a unique way
to continue the path. At broken vertices the path enters the vertex
through one bond and leaves it through another.
If the last bond $b$ added to the cluster points to a frozen vertex $v$,
the path bifurcates in the directions of the two outgoing arrows of $v$.
One of these directions can be considered as
belonging to the loop we came from, the other one as belonging
to a new loop. Since we also have to flip the second incoming
arrow of $v$, we are assured that this new loop also closes.
The two loops have to be flipped together.
In general, the zero-divergence condition
guarantees that all loops will eventually close.

%It is easy to see that our algorithm is correct.
%The proof of detailed balance is completely analogous to that for other
%cluster algorithms \cite{ClusterReviews,KandelDomany}. The main
%ingredient here is that $P_{\,\mbox{ul-lr}}$ and $P_{\,\mbox{ll-ur}}$
%already satisfy detailed balance locally.
%Furthermore, it is not difficult to see that any two allowed
%configurations can be connected by a finite number of cluster flips.
%Thus a finite power of the Markov matrix is ergodic.


We have now finished describing the procedure for constructing
clusters. In order to specify the algorithm completely, we must choose
values for the constants $q_i$, and decide how the clusters are flipped.
The former problem is of utmost importance, and it is the object of the
next chapter. For the cluster flips, we may use
both the Swendsen-Wang procedure and the Wolff single cluster flip
\cite{ClusterReviews}. We choose the latter, i.e.\ we
``grow'' a {\em single path} from a random starting bond $b_0$, and flip
it. The break-or-freeze decision is only needed for the vertices along
the path, so the computer time for one path is proportional to the
length of that path.

There are some distinct differences between our loop-clusters and more
conventional spin-clusters. For spin-clusters, the elementary objects
that can be flipped are spins; freezing binds them together into
clusters. Our closed loops on the other hand may be viewed as a part of
the {\em boundary} of spin-clusters
(notice that the boundary of spin clusters may
contain loops inside loops). It is reasonable to expect that in
typical cases, building a loop-cluster will cost less work than for a
spin-cluster. This is an intrinsic advantage of the loop algorithm.


\section{Optimization of free parameters}% \note{from 6V}

We have seen that freezing forces loops to be flipped together.
%\note{was commented:}
%
%(This effect is somewhat reminiscent of the problems caused by
%frustration in other cluster algorithms \cite{ClusterReviews}.)
%
Previous experience with cluster algorithms %\cite{ClusterReviews}
suggests that it is advantageous to be able to flip loops independently,
as far as possible.
We therefore introduce the principle of {\em minimal freezing}
as a guide for choosing the constants $q_i$:
we shall minimize the freezing probabilities,
given the constraints (\ref{KDnorm}) and $q_i \geq 0$.
%Furthermore, the larger a weight $\rho(u)$ is, the more we shall try to
%avoid its freezing.
In the next chapter we will show that for the case of the F~model, optimization
by minimal freezing does indeed minimize critical slowing down.
Here we discuss optimization for the 4 phases of the 6-vertex model,
usually denoted by capital roman numerals \cite{ReviewLieb}.

Let us first look at phase~IV, where $c > a+b$.
To minimize the freezing of weight $c$, we have to minimize $q_3$.
{}From (\ref{KDnorm}),  $q_3 = c-a-b + q_1+q_2+2q_6$.
With $q_i \geq 0$ this implies $q_{3,\mbox{\small min}} = c-a-b$.
The minimal value of $q_3$ can  only be chosen if {\em at the same time}
we set $q_1=q_2=0$, i.e.\ minimize (in this case do not allow for)
the freezing of the smaller weights $a$ and $b$.
The optimized parameters for phase~IV are then:
\begin{equation} \label{qFOUR}
\begin{array}{lll}
q_1=0, & q_2=0, & q_3=c-a-b, \\
q_4=b, & q_5=a, & q_6=0 \;.
\end{array}
\end{equation}
%

In phase I the situation is technically similar.
Here $a > b+c$, and we minimize freezing with $q_1=a-b-c$ and $q_2=q_3=0$.
The same holds for phase II, $b>a+c$, where we obtain minimal freezing
for $q_2=b-a-c$ and $q_1=q_3=0$.

Phase III (the massless phase) is characterized by
$a,b,c$ $< \half (a+b+c)$.
Here we can set all freezing probabilities to zero.
Thus,
\begin{equation} \label{qTHREE}
\begin{array}{lll}
q_1=0, & 2q_4=b+c-a \; , \\
q_2=0, & 2q_5=c+a-b \; , \\
q_3=0, & 2q_6=a+b-c \;.
\end{array}
\end{equation}




%***********************************************************************
 \section{Case of the F~model}
%***********************************************************************
%
The F~model is obtained from (\ref{qFOUR}) and (\ref{qTHREE})
as the special case $a=b=\exp{(-K)} \leq 1$, $c=1$.
It has a Kosterlitz-Thouless transition at $K_c=\ln 2$,
with a massless phase for $K\leq K_c$.

How do we choose the parameters $q_i$ here ?
Symmetry $a=b$ implies $q_1=q_2$ and $q_4=q_5$.
We can eliminate freezing of vertices $1,2,3,4,$ by setting
$q_1=q_2=0$. In (\ref{KDnorm}) this leaves one parameter, $q_3$:
%
\begin{equation}\label{Fmodelparameters}
\begin{array}{lll}
2q_4 &=& 1-q_3 \, , \\
2q_6 &=& \exp{(-K)} +q_3-1 \, .
\end{array}
\end{equation}
%
In the massless phase, we can minimize freezing by also setting $q_3=0$.
In the massive phase, $q_6\geq 0$ limits $q_3$.
Thus
\begin{equation}\label{qthree}
q_{3,\mbox{\small min}}= \left\{ \begin{array}{ll}
                                   1-2\exp{(-K)}, & K \geq K_c, \\
                                   0,             & K \leq K_c.
                        \end{array}
                \right.
\end{equation}
%
Notice that since $a=b$ in the F~model, the straight break-up,
the freezing of $a$, and that of $b$ are operationally the same thing.

If we choose $q_3=q_{3,\mbox{\small min}}$,
then for $K \myleq K_c$  vertices of type 5 and 6 are never frozen,
which has as a consequence that every path consists of a single loop.
This loop may intersect itself, like in the drawing of the figure ``8''.
%
For $K \mygt K_c$ on the other hand,
vertices of type 1, 2, 3 and 4 are never frozen, so we do not
continue a path along a straight line through any vertex.
As $K\rightarrow\infty$ (temperature goes to zero),
most vertices are of type 5 or 6, and they are almost always frozen.
Thus the algorithm basically flips between the two degenerate ground
states.

For the F~model we also have a spin-cluster
algorithm -- the VMR algorithm \cite{BCSOScluster}.
At $K=K_c$ and for $q_3=q_{3,\mbox{\small min}}$,
we  have a situation where the loop-clusters form
parts of the boundary of VMR spin-clusters. Since flipping a
loop-cluster is not the same as flipping a VMR cluster, we expect the
two algorithms to have different performances. We found (see
\cite{BCSOScluster} and the next section) that in
units of clusters, the VMR algorithm is more efficient, but
in work units, which are basically units of CPU time, the loop algorithm
wins. At $K_c/2$, where the loop-clusters are not at all related
to the boundary of VMR clusters,
we found the loop algorithm to be more efficient both in units of
clusters and in work units, with a larger advantage in the latter.


%***********************************************************************
 \section{F-model: Performance of the loop algorithm}
%***********************************************************************
%
We tested the loop algorithm on $L \times L$ square lattices
with periodic boundary conditions at two values of $K$:
at the transition point $K_c$, and
at $\kch$, which is deep inside the massless phase.
%
%(For equivalence with the model in \cite{BCSOScluster} we
% disallowed loops winding around the lattice).
%
We carefully analyzed autocorrelation functions
and determined the {\em exponential} autocorrelation time $\tau$.
At infinite correlation length, critical slowing down is quantified by
the relation (\ref{CSD}), $\tau\propto L^z$.

Local algorithms are slow, with $z \approx 2$.
For comparison, we performed runs with a local algorithm
that flips arrows around elementary plaquettes with
Metropolis probability, and indeed found $z = 2.2(2)$ at $K=K_c$.

In order to make sure that we do observe the slowest mode of the Markov
matrix, we measured a range of quantities and checked that they exhibit
the same $\tau$.
As in \cite{BCSOScluster}, it turned out that one can use quantities
defined on a sublattice in order to couple strongly to the slowest mode.
Specifically, we wrote the energy as a sum over two sublattice quantities.
We shall present more details of this phenomenon elsewhere.
Let us however note here that for the total energy,
%The two sublattice energies\ %
%\cite{BCSOScluster}
%add up to the total energy. The constraints of the model
%cause them to be strongly anticorrelated.
%Within our precision the
the true value of $\tau$ was not visible within our precision
%total energy, which decay very quickly. Only for the largest lattices do
except for a weak hint of a long tail in the autocorrelations on the
largest lattices we considered.
%
%A similar situation occurred in \cite{BCSOScluster}, where,
%when decreasing the statistical errors,
%the decay governed by the true $\tau$ eventually became visible.
Note that as a consequence of this situation,
the so-called ``integrated  autocorrelation time''
\cite{ClusterReviews} is much smaller than $\tau$, and it would be
completely misleading to evaluate the algorithm based only on its values.
%
%\note{e.g. Polarization: sharp turn in autocorr.fct. on large lattices}

We shall quote autocorrelation times $\tau$ in units of
``sweeps'' \cite{ClusterReviews}, defined such that on the average each
bond is updated once during a sweep.
Thus, if $\taucl$ is the autocorrelation time in units of clusters%
%@ (including those that were rejected because they wound around the lattice)%
, then
%
$\tau = \taucl \times
       \langle\mbox{cluster size}\rangle / {(2L^2)}$.
%
Each of our runs consisted of between 50000 and 200000 sweeps.
%
Let us also define $\zcl$ by
$\taucl \propto L^{\mbox{\protect\small $\zcl$}}$,
and a cluster size exponent $c$ by
$\langle\mbox{cluster size}\rangle \propto L^{\mbox{\protect\small $c$}}$.
We then have:
\begin{equation}\label{taucl}
   z = \zcl - (2-c) \, .
\end{equation}

%%%%%%%%%%%% TABLE tabtau %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{table}[tb]
 \centering
\vskip2ex
\begin{tabular}{|r|r|r|}
\hline\str
 $L$ & \mc{1}{c|}{$K=K_c$} & \mc{1}{c|}{$K=\kch$}
\\[.5ex]\hline \str
%%%%%%%%% SHORT TABLE WITH HG's NORMALIZATION,
  8 & 1.8(1)   & 4.9(4) \\
 16 & 3.0(2)   & 5.6(2) \\
 32 & 4.9(4)   & 6.2(3) \\
 64 & 7.2(7)   & 7.4(3) \\
128 &15.5(1.5) & 8.3(2) \\
256 &20.5(2.0) &
%
\tabhline
 $z$  & \zatKc  &\zatKch
\\[.3ex] \hline
\end{tabular}
%
 \caption[dummy]{\label{tabtau} \parbox[t]{.85\textwidth}{
                              Exponential autocorrelation time $\tau$
                              at $q_3 \!=\! \qmin$,
                              and the resulting \mbox{dynamical}
                              critical exponent $z$.
                          }}%%
%
\end{table}
%%

Table 2 %\ref{tabtau}
shows the autocorrelation time $\tau$ for the optimal choice
$q_3\myeq q_{3,\mbox{\small min}}$.
At $K=\kch$, deep inside the massless phase, critical slowing down is almost
completely absent. A fit according to eq.\ \ref{CSD} gives $z=\zatKch$.
The data are also consistent with $z=0$ and only logarithmic growth.
For the cluster size exponent $c$
we obtained $c=1.446(2)$, which points to
the clusters being quite fractal (notice that $\zcl=0.74(2)$).
At the phase transition $K = K_c$ we obtained
$z=\zatKc$, which is still small.
The clusters seem to be less fractal: $c=1.060(2)$,
so that $\zcl=1.65(5)$.

We noted above that at
$K=K_c$ and for the optimal choice of $q_3$, the loop-clusters are related
to the VMR spin-clusters. In \cite{BCSOScluster} we obtained for
the VMR algorithm at $K=K_c$ the result $\zcl=1.22(2)$, but we
had $c=1.985(4)$, which left us with $z=1.20(2)$. In this case, although
in units of clusters the VMR algorithm is more efficient,
the smaller dimensionality of the loop-clusters more than make
up for this, and in CPU time the loop algorithm is more efficient.

As mentioned, no critical slowing down is visible for the integrated
autocorrelation time $\tauint(E)$ of the total energy.
At $K=K_c$, $\tauint(E)$ is only 0.80(2) on the largest lattice,
and we find the dynamical exponent $\zint(E) \approx 0.20(2)$.
At $K=\kch$, $\tauint(E)$ is $1.1(1)$ on all lattice sizes,
so $\zint(E)$ is zero.

What happens for non-minimal values of $q_3\,$?
Table 3 %\ref{tabz}
shows our results on the dependence of $z$ on $q_3$.
$z$ rapidly increases as $q_3$ moves away from $q_{3,\mbox{\small min}}$.
This effect seems to be stronger at $\kch$ than at $K_c$.
We thus see that the optimal value of
$q_3$ indeed produces the best results, as conjectured
{}from our principle of {\em least possible freezing}.

%%%%%%%%%%%%%%% TABLE tabz %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\begin{table}[tb]  \centering
\begin{tabular}{|c|c|r|}
\hline\str
 $K$   & $q_3$ & \mc{1}{c|}{$z$}
        \\[.5ex] \hline  \str
%
$\kch$ & 0    &$ \zatKch$    \\[.3ex]
$\kch$ & 0.10 &$ 1.90(5)$    \\[.3ex]
$\kch$ & 0.20 &$\geq2.6(4)$
\tabhline
$K_c$  & 0     &$ \zatKc $    \\[.3ex]
$K_c$  & 0.05  &$ 0.77(6)$    \\[.3ex]
$K_c$  & 0.10  &$ 0.99(6)$    \\[.3ex]
$K_c$  & 0.20  &$\geq2.2(1)$
\\[.3ex] \hline
\end{tabular}
%%
 \caption[dummy]{\label{tabz}\parbox[t]{.85\textwidth}{
       Dependence of the dynamical critical exponent
       $z$ on the parameter $q_3$.
       We use ``$\geq$'' where for our lattice sizes $\tau$
       increases faster than a power of $L$.
   }}%%
%
\end{table}
%%%%%%%%%%%%%%%%%%%%%%%% END OF TABLE tabz %%%%%%%%%%%%%%%%%

In the massive phase close to $K_c$,
we expect that
$z(K_c)$ will determine the behaviour of $\tau$
in a similar way as in ref.\ \cite{BCSOScluster}.
To confirm this, a finite size scaling analysis of
$\tau$ is required.
In order to complete the study of the loop algorithm's performance,
we should also investigate it at the KDV transition.

In summary, the loop algorithm strongly reduces critical slowing down,
{}from $z=2.2(2)$ for the Metropolis algorithm,
down to $z=\zatKc$ at $K_c$ and $z=\zatKch$ at $\kch$ deep inside the
massless phase.
For the integrated autocorrelation time of the total energy,
no critical slowing down is visible in either case.
\pagebreak %%@%
%***********************************************************************
 \section{Quantum Spin Systems}
%***********************************************************************

Particularly promising is the possibility of
{\em accelerating Quantum Monte Carlo simulations}, since
quantum spin systems in one and two dimensions can be mapped into
vertex models in $1+1$ and $2+1$ dimensions via the Trotter formula
and suitable splittings of the Hamiltonian \cite{QMC}.
%
The simplest example is the spin $\half$ $xxz$ quantum chain, which is
mapped directly into the 6-vertex model.
For higher spins, more complicated
vertex models result (e.g.\ 19-vertex model for spin one).

For $(2+1)$ dimensions, different splittings of the Hamiltonian
lead to quite different vertex models, in particular on quite
different lattice types \cite{QMC}. For example, in
the case of spin $\half$ we can choose between 6-vertex models
on a quite complicated $2+1$ dimensional lattice, and models on a
{\em bcc} lattice, with 8 bonds and a large number of configurations
per vertex.

For the simulation of the {\em 2-dimensional Heisenberg antiferromagnet
and ferromagnet} using the former splitting,
all relevant formulas have been worked out
in the present paper.
Actually, the low temperature properties of the antiferromagnet
have recently been investigated by Wiese and Ying
\cite{WieseHeisenberg} {\em using our algorithm}. Their calculation is,
in our opinion, the first high quality verification of the magnon
picture for the low lying excitations.
In particular, this excludes to a much higher degree of confidence
than before the speculation (some years ago widespread) that the
model had a nonzero mass gap.

Notice that, similarly to other cluster algorithms \cite{ClusterReviews},
it is straightforward to define improved observables. The investigation
\cite{WieseHeisenberg} in fact uses them.

Let us also remark that
the loop algorithm can easily change global properties
like the number of world lines or the winding number (see \cite{QMC}).
Thus it is well suited for simulations in the grand canonical ensemble.
%
Last, but not least, the loop algorithm also opens up a new avenue
for taming the notorious fermion sign problem  \cite{Signpaper}.

%
%There is a paper  by Wiese and Ying \cite{WieseYing}
%on a different cluster algorithm for spin $\half$ quantum spin systems.
%After mapping to a vertex model
%similar to the one we refer to,
%they combine vertices into ``block-spins''  which are then used
%in a standard spin-cluster construction. This approach
%restricts the possible updates of the arrows.
%In our language, their clusters are sets of loops that are frozen together,
%i.e., that have to be flipped together.
%For some interesting cases, e.g., the one-dimensional
%Heisenberg ferromagnet
%and two-dimensional Heisenberg ferromagnet and anti-ferromagnet,
%the additional freezing leads to the problem of
%{\em frustration} for the block-spin clusters.
%We expect our algorithm to perform
%better in these cases, both because our clusters have less loops and
%because of the added flexibility offered by the
%possibility to optimize.
%

%***********************************************************************
\section{Conclusions}
%***********************************************************************
%
We have presented a new type of cluster algorithm.
It flips closed paths of bonds in vertex models.
Constraints are automatically satisfied.
We have successfully tested our algorithm for the F~model and found
remarkably small dynamical critical exponents.

There are many promising and
straightforward applications of our approach,
to other vertex models, and to 1+1 and 2+1 dimensional
quantum spin systems.
Investigations of such systems are in progress.
In particular, we believe that our generalizations of the
freeze-delete scheme can be adapted for other models like the
8-vertex model.

Already, the loop algorithm has found important applications
in the study of the 2-dimensional Heisenberg antiferromagnet.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%     ACKNOWLEDGEMENTS
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section*{Acknowledgements}

This work was supported in part by the Ger\-man-Israeli
Foundation for Research and Development (GIF) and
by the Basic Research Foundation of
the Israel Academy of Sciences and Humanities.
We would like to express our gratitude
to the HLRZ at KFA J\"ulich % where most of our computer runs were performed.
and to the DOE for providing the necessary computer time
for the F~model study.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%     BIBLIOGRAPHY
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
   \pagebreak%%[3]
   
%
\end{document}
%
%        1         2         3         4         5         6         7
%23456789012345678901234567890123456789012345678901234567890123456789012345678

